Author's response to reviews

Title: Association between smoking and chronic kidney disease: a case control study

Authors:

Rabi Yacoub (Rabiyaco@buffalo.edu)
Habib Habib (Hhabib@smmcnj.org)
Ayham Lahdo (Ayham_lahdo@hotmail.com)
Radwan Al Ali (Radgali@gmail.com)
Leon Varjabetian (leonvarj@buffalo.edu)
George Atalla (Gaa82@hotmail.com)
Nader Kassiss Akl (Naderkassiss@yahoo.com)
Salim Dakhil (Salimdakhil@yahoo.com)
Saeed Alahdab (Saeedalahdab@yahoo.com)
Sami Albitar (Samiabitar@hotmail.com)

Version: 4 Date: 24 November 2010

Author’s response to reviews:

Major revisions:
Affiliations: updated.
Minor revisions:
Title: corrected
Highlighting/tracking: removed.
Authors’ contributions: in paragraph.
Tables: title above the table.